Author's response to reviews

Title: Investigating the efficacy of integrated cognitive behavioral therapy for adult treatment seeking substance use disorder patients with comorbid ADHD: Study protocol of a randomized controlled trial.

Authors:

Katelijne van Emmerik-van Oortmerssen (katelijnevo@hotmail.com)
Ellen Vedel (ellen.vedel@jellinek.nl)
Maarten W Koeter (m.w.koeter@amc.uva.nl)
Kim de Bruijn (kim.de.bruijn@jellinek.nl)
Jack J.M. Dekker (jack.dekker@arkin.nl)
Wim van den Brink (w.vandenbrink@amc.uva.nl)
Robert A Schoevers (r.a.schoevers@umcg.nl)

Version: 2 Date: 6 February 2013

Author's response to reviews:

BMC Psychiatry
Section Editor
Amsterdam, February 6th, 2013

Dear editor,

Herewith, we submit a revised version of our manuscript entitled ‘Investigating the efficacy of integrated cognitive behavioral therapy for adult treatment seeking substance use disorder patients with comorbid ADHD: Study protocol of a randomized controlled trial’ to BMC Psychiatry for review. In this version, we included proofs of the following:

1. We received external funding of 130.000 euro by a Dutch Foundation (Fonds NutsOhra); the original document is uploaded, with a signed translated version of the relevant part.

2. The study protocol has been approved by the local review board of the Academic Medical Centre of the University of Amsterdam. The original document is attached, followed by a signed translated version of the relevant part.

Currently, there are no other papers on this study submitted.

Yours sincerely, on behalf of all authors,

Katelijne van Emmerik–van Oortmerssen, M.D.
Arkin Mental Health and Addiction, Research Department
Klaprozenweg 111
1033 NN Amsterdam, The Netherlands
Tel: +31 6 25 465 991
Email: katalijne.van.oortmerssen@arkin.nl